Author's response to reviews

Title: The impact of image dynamic range on texture classification of brain white matter.

Authors:

Doaa Ghoneim (dmahmoud@uaeu.ac.ae)
Mariam K Alkaabi (dmahmoud@uaeu.ac.ae)
Jacques D de Certaines (jd.de-certaines@wanadoo.fr)
Frank Goettsche (frank.goettsche@imk.fzk.de)

Version: 7 Date: 22 September 2008

Author's response to reviews: see over
Dear Sir/ Madame

Please find attached the revised manuscript after taking into account the referees suggestions and comments. Kindly find the following highlighted in the newly revised manuscript:

1. Dictionary revisions (highlighted in yellow).

2. Scientific revisions: ROC curve was analyzed using the software SPSS, and AUC calculated for each test and included in the results, discussion and conclusion (highlighted in blue). Table 1 was updated with the AUC values that correspond to each test. A new figure was also added showing example of two ROC curves corresponding to the highest and lowest AUC in the study, according to the suggestions of the referees.

Finally I should sincerely thank you and the referees for the high quality of the reviewing process.

If you still need more information or have further comments please do not hesitate to contact me.